Author's response to reviews

Title: Evolving health information technology and the timely availability of visit diagnoses from ambulatory visits: A natural experiment in an integrated delivery system

Authors:

Naomi S Bardach (bardachn@peds.ucsf.edu)
Jie Huang (jie.huang@nsmtp kp.org)
Richard Brand (rbrand@epi.ucsf.edu)
John Hsu (John.T.Hsu@nsmtp kp.org)

Version: 5 Date: 15 July 2009

Author's response to reviews:

The small revisions we have made are as follows:
Correction of two references
Change EHR to EMR in Table 1
Addition of abbreviation legend at bottom of Table 1